Author's response to reviews

Title: An evaluation of classification systems for stillbirth

Authors:

Vicki Flenady (Vicki.flenady@mater.org.au)
J. Frederik Froen (Frederik.Froen@fhi.no)
Halit Pinar (halit.pinar@gmail.com)
Rozbeh Torabi (rtorabi@yahoo.com)
Eli Saastad (eli.saastad@fhi.no)
Grace Guyon (Grace.Guyon@capitalhealth.ca)
Laurie Russell (Laurie.Russell@capitalhealth.ca)
Adrian Charles (Adrian.Charles@health.wa.gov.au)
Catherine Harrison (Catherine.harrison@health.wa.gov.au)
Lawrence Chauke (chaukehl@telkomsa.net)
Robert Pattinson (robert.pattinson@up.ac.za)
Rachel Koshy (rachelkphilip@hotmail.com)
Safiah Bahrin (safiah@moh.gov.my)
Glenn Gardener (Glenn.Gardener@mater.org.au)
Katie Day (Katie.Day@mater.org.au)
Karin Petersson (Karin.petersson@karolinska.se)
Adrienne Gordon (adrienne.gordon@email.cs.nsw.gov.au)
Kristen Gilshenan (Kristen.Gilshenan@mater.org.au)

Version: 11 Date: 11 June 2009

Author’s response to reviews:

11.06.09

Dear BioMed Central

Thank you for additional comments on the formatting of the manuscript. The following have been amended:

Affiliations - Each affiliation should be written in full, in the following format: department, institute, city, and country. This may mean repeating institutional addresses if the authors are from different departments within an institution.

Authors’ contributions - Please indicate that all authors read and approved the final manuscript.

Reference 8 - please provide full and accurate details for this reference. This means for an Article providing year/volume/issue/page numbers and for a book the publisher/city/year.
Reference 18 - please provide full and accurate details for this reference. This means for an Article providing year/volume/issue/page numbers and for a book the publisher/city/year.

Reference 67, In press - Has this article now been published? If so, please provide the citation details

Unfortunately, we are unable to identify the amendment required for reference 45 ( below). We are not able to provide a weblink (if this is what is required) as one does not exist. Please let me know of any other changes need to be made.

Reference 45 - please provide full and accurate details for this reference. This means for an Article providing year/volume/issue/page numbers and for a book the publisher/city/year.

Thank you for your assistance

Kind regards

Vicki Flenady